Author's response to reviews

Title: Fluorophotometry as a diagnostic tool for the evaluation of dry eye disease

Authors:

Magid M Fahim (magid2000@hotmail.com)
Shamim A Haji (shamimhaji@yahoo.com)
Chakravarthy V Koonapareddy (chak.reddy@mssm.edu)
Vincent C Fan (vincent.fan@mssm.edu)
Penny A Asbell (penny.asbell@mssm.edu)

Version: 5 Date: 5 May 2006

Author's response to reviews: see over
Formatting Changes:

05 May 2006

1. Title and Abstract corrected and identical in manuscript and submission system.
   - Methods:
     - TM removed and replaced with TradeMark
     - “installed” corrected with “instilled”
   - Results:
     - “installation” replaced with “instillation”
     - “installation” replaced with “instillation”

2. Authors: updated additional author; and authors inputed into correct order in submission system.

3. Affiliations: section added on Title pages for each author and country added. FAX updated.

4. Qualifications: removed from each author on title page.

5. Keywords section removed.

6. Section headings replaced with non-all-capsitals.

7. “Introduction” replaced with “Background”

8. References throughout manuscript replaced with brackets and non-superscripts numbers.

9. Author’s Contributions reordered in author listed format and updated.

10. References: all author names included and et al removed.

11. Tables: gray removed from cells and word “figure #1, 2” removed.
    (new edited figure files uploaded)